Author's response to reviews

Title: Risk factors for suicide in Hungary: a case-control study

Authors:

Kitty Almasi (almasie@externet.hu)
Nora Belso (belsonora@axelero.hu)
Navneet Kapur (nav.kapur@manchester.ac.uk)
Roger Webb (roger.webb@manchester.ac.uk)
Jayne Cooper (jayne.b.cooper@manchester.ac.uk)
Sarah Hadley (sarah.hadley@manchester.ac.uk)
Michael Kerfoot (michael.kerfoot@manchester.ac.uk)
Graham Dunn (graham.dunn@manchester.ac.uk)
Peter Sotonyi (sotpet@igaz.sote.hu)
Zoltan Rihmer (rihmerz@kut.sote.hu)
Louis Appleby (louis.appleby@manchester.ac.uk)

Version: 4 Date: 2 June 2009

Author's response to reviews:

Derar Dr Alam,

We have made the formatting changes requested.

Specifically, we have restructured the abstract, have inserted a conclusion section and cited tables individually rather than in ranges.

With respect to the minor revisions, we have used sentence case for the title, removed the qualifications, used superscript for affiliations, included email addresses for all authors on the title page, removed the keywords, renamed the introduction and methods sections, removed the highlighting and underlining.

Many thanks
Best wishes
Nav Kapur

Prof N Kapur on behalf of the authors